Author's response to reviews

Title: Pulmonary artery intimal sarcoma a case report

Authors:

Hongqing HQ Long (longhongqing@163.com)
Qin Q QIN (luyin12340613@hotmail.com)
Conghua CH Xie (chxie-65@hotmail.com)

Version: 4 Date: 4 January 2008

Author's response to reviews: see over
According to the comments of reviewer “Antonella Coli”, we made the following revisions:

1. We added the word ‘local’ in the par.2 line 1 to make it clear that the patient had a local resection which means uncompletely.
2. We deleted the Ref 1 in par.4 line5, as the reviewer suggested that Ref 1 has nothing to do with radiation therapy. And changed the “patients” to “patient”
3. We did some revision of the reference 2, 5 and 9.
4. As to the first point of this reviewer, I think it is clear that “no abnormalities were detected in abdomen” was in the physical examination, so we did not revise it.

According to the comments of reviewer “Maithili Rao”, we made the following revisions:

1. We deleted some sentences in par.1 of case presentation, so as to shorten the detailed technical explanations of the ECHO and CT findings.
2. We added arrows in the figure 1 and 3 to indicate the areas of interest.

Also, we did some language corrections to make sure it is more intelligible to readers.